Author’s response to reviews

Title: A 3-arm Randomized Controlled Trial on the Effects of Dance Movement Intervention and Exercises on Elderly with Early Dementia

Authors:

Rainbow Ho (tinho@hku.hk)
Jacob Cheung (cheungjk@hku.hk)
Wai Chi Chan (waicchan@hku.hk)
Irene Cheung (ireneh@hku.hk)
Linda Lam (cwlam@cuhk.edu.hk)

Version: 1 Date: 12 Aug 2015

Author’s response to reviews:

Dear Editor,

The following changes were made in response to your suggestions;

1. Email have been added to the first page.

2. RESEARCH OBJECTIVES is reformatted to be part of background.

3. METHODS/PLAN is renamed to Methods/Design

4. DATA ANALYSIS is reformatted as a sub-section of Methods/Design

5. Sentence is changed to convey ethics has already been approved and the name of the ethics committee are included in the text.

6. Flow diagram are formatted following the CONSORT guidelines and uploaded separately in the system. A Figure legend is also included after references. A sentence regarding the adherence to CONSORT guidelines is also included in the manuscript under "Acknowledgements".

Thank you for your help and attention.